Author’s response to reviews

Title: Cefaclor as a first-line treatment for acute uncomplicated cystitis

Authors:

Dai Koguchi (ro-calcio-kean-di-daissimo@hotmail.co.jp)
Yasukiyo Murakami (yaskiyomura@yahoo.co.jp)
Masaomi Ikeda (ikeda.masaomi@grape.plala.or.jp)
Masato Dobashi (mdbashi0103@gmail.com)
Junichiro Ishii (jun-ro@dc4.so-net.ne.jp)

Version: 2 Date: 20 Mar 2020

Author’s response to reviews:

March 20, 2020
Dr. Hayley Henderson and Dr. Doris Cuckovic
Editors, BMC Urology

RE: Mn.No. BURO-D-20-00005R2
Title: "Cefaclor as a first-line treatment for acute uncomplicated cystitis"

Dr. Hayley Henderson and Dr. Doris Cuckovic

We deeply appreciate your kindness and have reviewed the comments from the editor and made revisions. A list of the response to each of the comments follows.

Editor Comments:

1. Body of Manuscript:
   a. Please provide the "Conclusion" heading
   b. Please rename Patients and Methods to "Methods" only

2. DECLARATIONS SECTION: Please re-arrange the Declarations sub-headings accordingly.

Answer: We corrected what the editor pointed out accordingly in the attached document. Again, thank you very much for your kindness.